Author’s response to reviews

Title: Intravenous Non-high-dose Pantoprazole is equally effective as High-dose Pantoprazole in Preventing Rebleeding among Low Risk Patients with a Bleeding Peptic Ulcer after Initial Endoscopic Hemostasis

Authors:

Chih-Ming Liang (gimy54861439@gmail.com)
Jyong-Hong Lee (lobby.lee@gmail.com)
Yuan-Hung Kuo (0104kuo@pchome.com.tw)
Keng-Liang Wu (kengliang_wu@yahoo.com.tw)
Yi-Chun Chiu (chiuku@ms14.hinet.net)
Yeh-Pin Chou (yehpin@yahoo.com)
Ming-Luen Hu (allenhu7296@yahoo.com.tw)
Wei-Chen Tai (luketai1019@gmail.com)
King-Wah Chiu (c471026@ms6.hinet.net)
Tsung-Hui Hu (hutsh@ms32.hinet.net)
Seng-Kee Chuah (chuahsk@seed.net.tw)

Version: 2 Date: 6 March 2012

Dear Editor:

We would like to submit the revised manuscript (MS: 1975264622645304-R1) entitled “Intravenous Non-high-dose Pantoprazole is equally effective as High-dose Pantoprazole in Preventing Rebleeding among Low Risk Patients with a Bleeding Peptic Ulcer after Initial Endoscopic Hemostasis” for your consideration of publication in BMC Gastroenterology.

Thank you very much for the opportunity to revise the manuscript. We have made extensive changes and answered point to point queries of the reviewers. English has been revised and re-edited by the editing company (proof is also submitted).

We look forward to hearing from your further comments soon. Thank you very much!

Best regards,
Yours sincerely,
Dr. Seng-Kee Chuah